Author's response to reviews

Title: Age-specific prevalence of the metabolic syndrome defined by the International Diabetes Federation and the National Cholesterol Education Program: the Norwegian HUNT 2 study

Authors:

Bjorn Hildrum (bjorn.hildrum@hnt.no)
Arnstein Mykletun (arnstein.mykletun@uib.no)
Torstein Hole (torstein.hole@helse-sunnmore.no)
Kristian Midthjell (kristian.midthjell@ntnu.no)
Alv A Dahl (alvd@ulrik.uio.no)

Version: 6 Date: 21 August 2007

Author's response to reviews:

The BioMed Central Editorial Production Team

Object: MS: 1001909799120644

Dear Editor,

Thank you for e-mail dated August 16, 2007.

We have made the requested formatting changes and some other minor corrections in the manuscript.

Best regards (on behalf of the authors)

Bjorn Hildrum, MD